%%% subcubes.tex
%%% 
%%% by Blair Seidler, Svante Janson, and Doron Zeilberger (xx pages)
%%% a LaTeX file

\documentclass{PJM}
%\usepackage{amsmath}
%\usepackage{amssymb}
%\usepackage{hyperref}
%\usepackage[latin1]{inputenc}   om svenska bokst}ver
%\usepackage{upref}
%\usepackage[notcite,notref]{showkeys} % shows labels
%\usepackage{natbib}
%\bibpunct[, ]{[}{]}{;}{n}{,}{,}

%\addtolength{\textwidth}{10mm}

%\linespread{1.6}  % double spaced

%\numberwithin{equation}{section}

%\setcounter{secnumdepth}{1}

\allowdisplaybreaks


%%%%% THEOREMS ETC %%%%%


%\newtheorem{theorem}{Theorem}[section]
%\newtheorem{lemma}[theorem]{Lemma}
%\newtheorem{proposition}[theorem]{Proposition}
%\newtheorem{corollary}[theorem]{Corollary}
%\newtheorem{conj}[theorem]{Conjecture}
%\newtheorem{Alg}[theorem]{Algorithm}

%\theoremstyle{definition}
%\newtheorem{example}[theorem]{Example}
%\newtheorem*{definition}{Definition}
%\newtheorem{problem}[theorem]{Problem}
%\newtheorem{remark}[theorem]{Remark}
%\newtheorem*{personal}{Personal remark}
\newtheorem{tom}{}
\newtheorem*{acks}{Acknowledgements}
\newtheorem*{ack}{Acknowledgement}

\theoremstyle{remark}

%\newenvironment{acks}{\section*{Acknowledgement}}{}
%\newenvironment{algo}{\begin{enumerate}}{\end{enumerate}}

\newenvironment{romenumerate}[1][0pt]{% optional argument changes indentation
\addtolength{\leftmargini}{#1}\begin{enumerate}% gives (i), (ii) etc.
 \renewcommand{\labelenumi}{\textup{(\roman{enumi})}}%
 \renewcommand{\theenumi}{\textup{(\roman{enumi})}}%
 }{\end{enumerate}}

\newcounter{oldenumi}
\newenvironment{romenumerateq}% continues numbering from previous romenumerate
{\setcounter{oldenumi}{\value{enumi}}
\begin{romenumerate} \setcounter{enumi}{\value{oldenumi}}}
{\end{romenumerate}}



% thmenumerate gives (i) run in after ``Theorem ...''
% now works with \label too, but needs '%' after \label{...}%
\newcounter{thmenumerate}
\newenvironment{thmenumerate}
{\setcounter{thmenumerate}{0}%
 \renewcommand{\thethmenumerate}{\textup{(\roman{thmenumerate})}}%
 \def\item{\par% \ifnum\thethmenumerate=0\else\par\fi %\noindent\fi
 \refstepcounter{thmenumerate}\textup{(\roman{thmenumerate})\enspace}}
}
{}

\newcounter{romxenumerate}   %less indented than standard.
\newenvironment{romxenumerate}
  {\begin{list}
    {\upshape(\roman{romxenumerate})}
    {\addtolength{\leftmargin}{-10pt} %ad lib 
     \addtolength{\labelwidth}{-10pt}
%     \addtolength{\labelsep}{0pt}
     %\setlength{\rightmargin}{0pt}
%\def\makelabel##1{K##1}
 \renewcommand{\theromxenumerate}{\textup{(\roman{romxenumerate})}}
     \usecounter{romxenumerate}} 
% \renewcommand{\theromxenumerate}{\textup{(\roman{romxenumerate})}}
}
  {\end{list}}


\newcounter{xenumerate}   %no left indentation; thus wider lines
\newenvironment{xenumerate}
  {\begin{list}
    {\upshape(\roman{xenumerate})}
    {\setlength{\leftmargin}{0pt}
     \setlength{\rightmargin}{0pt}
     \setlength{\labelwidth}{0pt}
     \setlength{\itemindent}{\labelsep}
     \setlength{\topsep}{0pt}
     \usecounter{xenumerate}} }
  {\end{list}}

\newcommand\pfitem[1]{\par(#1):}

\newcommand\step[2]{\smallskip\noindent\emph{Step #1: #2} \noindent}


\newcommand{\refT}[1]{Theorem~\ref{#1}}
\newcommand{\refC}[1]{Corollary~\ref{#1}}
\newcommand{\refL}[1]{Lemma~\ref{#1}}
\newcommand{\refR}[1]{Remark~\ref{#1}}
\newcommand{\refS}[1]{Section~\ref{#1}}
\newcommand{\refSS}[1]{Subsection~\ref{#1}}
%\newcommand{\refP}[1]{Proposition~\ref{#1}}
\newcommand{\refP}[1]{Problem~\ref{#1}}
\newcommand{\refD}[1]{Definition~\ref{#1}}
\newcommand{\refE}[1]{Example~\ref{#1}}
\newcommand{\refF}[1]{Figure~\ref{#1}}
\newcommand{\refApp}[1]{Appendix~\ref{#1}}
\newcommand{\refConj}[1]{Conjecture~\ref{#1}}
\newcommand{\refTab}[1]{Table~\ref{#1}}
\newcommand{\refand}[2]{\ref{#1} and~\ref{#2}}


% Marginal notes and labelling macros
\newcommand\marginal[1]{\marginpar{\raggedright\parindent=0pt\tiny #1}}
\newcommand\SJ{\marginal{SJ}}
\newcommand\SJm[1]{\marginal{SJ: #1}}
%\newcommand\kolla{\marginal{CHECK! SJ}}
\newcommand\kolla{\marginal{KOLLA!}}

\newcommand\linebreakx{\unskip\marginal{$\backslash$linebreak}\linebreak}

%\def\nomarginal{%
% \def\marginal##1{\ignorespaces}%
% \overfullrule=0pt \message{*** no marginals or black boxes ***}}


\begingroup
  \count255=\time
  \divide\count255 by 60
  \count1=\count255
  \multiply\count255 by -60
  \advance\count255 by \time
  \ifnum \count255 < 10 \xdef\klockan{\the\count1.0\the\count255}
  \else\xdef\klockan{\the\count1.\the\count255}\fi
\endgroup

% further AmSTeX adaptions
\newcommand\nopf{\qed}   % for theorem without proof
\newcommand\noqed{\renewcommand{\qed}{}} % for proof with explicit \qed
\newcommand\qedtag{\tag*{\qedsymbol}}



%\DeclareMathOperator*{\sumsum}{\sum\sum}
\DeclareMathOperator*{\sumx}{\sum\nolimits^{*}}
\DeclareMathOperator*{\sumxx}{\sum\nolimits^{**}}
\DeclareMathOperator*{\sumy}{\sum\nolimits^{*}}
\DeclareMathOperator*{\sumyq}{\sum\nolimits^{*\!\!\!}}
\newcommand{\sumi}{\sum_{i=0}^\infty}
\newcommand{\sumj}{\sum_{j=0}^\infty}
\newcommand{\sumk}{\sum_{k=0}^\infty}
\newcommand{\summ}{\sum_{m=0}^\infty}
\newcommand{\sumn}{\sum_{n=0}^\infty}

\newcommand\set[1]{\ensuremath{\{#1\}}}
\newcommand\bigset[1]{\ensuremath{\bigl\{#1\bigr\}}}
\newcommand\Bigset[1]{\ensuremath{\Bigl\{#1\Bigr\}}}
\newcommand\biggset[1]{\ensuremath{\biggl\{#1\biggr\}}}
\newcommand\xpar[1]{(#1)}
\newcommand\bigpar[1]{\bigl(#1\bigr)}
\newcommand\Bigpar[1]{\Bigl(#1\Bigr)}
\newcommand\biggpar[1]{\biggl(#1\biggr)}
\newcommand\lrpar[1]{\left(#1\right)}
\newcommand\bigsqpar[1]{\bigl[#1\bigr]}
\newcommand\Bigsqpar[1]{\Bigl[#1\Bigr]}
\newcommand\biggsqpar[1]{\biggl[#1\biggr]}
\newcommand\lrsqpar[1]{\left[#1\right]}
\newcommand\xcpar[1]{\{#1\}}
\newcommand\bigcpar[1]{\bigl\{#1\bigr\}}
\newcommand\Bigcpar[1]{\Bigl\{#1\Bigr\}}
\newcommand\biggcpar[1]{\biggl\{#1\biggr\}}
\newcommand\lrcpar[1]{\left\{#1\right\}}
\newcommand\abs[1]{|#1|}
\newcommand\bigabs[1]{\bigl|#1\bigr|}
\newcommand\Bigabs[1]{\Bigl|#1\Bigr|}
\newcommand\biggabs[1]{\biggl|#1\biggr|}
\newcommand\lrabs[1]{\left|#1\right|}
\def\rompar(#1){\textup(#1\textup)}    % usage: \rompar(...)
\newcommand\xfrac[2]{#1/#2}
\newcommand\xpfrac[2]{(#1)/#2}
\newcommand\parfrac[2]{\lrpar{\frac{#1}{#2}}}
\newcommand\bigparfrac[2]{\bigpar{\frac{#1}{#2}}}
\newcommand\Bigparfrac[2]{\Bigpar{\frac{#1}{#2}}}
\newcommand\biggparfrac[2]{\biggpar{\frac{#1}{#2}}}
\newcommand\xparfrac[2]{\xpar{\xfrac{#1}{#2}}}
\newcommand\expX[1]{\exp\Bigl(#1\Bigr)}
\newcommand\expx[1]{\exp\bigl(#1\bigr)}
\newcommand\expQ[1]{e^{#1}}
\def\xexp(#1){e^{#1}}
\newcommand\ceil[1]{\lceil#1\rceil}
\newcommand\floor[1]{\lfloor#1\rfloor}
\newcommand\fract[1]{\{#1\}}
%\newcommand\setx[1]{\set{1,\dots,#1}}
\newcommand\setn{\set{1,\dots,n}}
\newcommand\ntoo{\ensuremath{{n\to\infty}}}
\newcommand\Ntoo{\ensuremath{{N\to\infty}}}
\newcommand\asntoo{\text{as }\ntoo}
\newcommand\ktoo{\ensuremath{{k\to\infty}}}
\newcommand\stoo{\ensuremath{{s\to\infty}}}
\newcommand\bmin{\wedge}
\newcommand\norm[1]{\|#1\|}
\newcommand\bignorm[1]{\bigl\|#1\bigr\|}
\newcommand\Bignorm[1]{\Bigl\|#1\Bigr\|}
\newcommand\downto{\searrow}
\newcommand\upto{\nearrow}
%\newcommand\upto{\uparrow}
\newcommand\half{\tfrac12}
\newcommand\thalf{\tfrac12}

\newcommand\iid{i.i.d.\spacefactor=1000}    
\newcommand\ie{i.e.\spacefactor=1000}
\newcommand\eg{e.g.\spacefactor=1000}
\newcommand\viz{viz.\spacefactor=1000}
\newcommand\cf{cf.\spacefactor=1000}
\newcommand{\as}{a.s.\spacefactor=1000}
\newcommand{\aex}{a.e.\spacefactor=1000}
\renewcommand{\ae}{\vu}  % som felkoll
%\newcommand\whp{\textbf{whp}}
\newcommand\whp{{w.h.p.\spacefactor=1000}}

\newcommand\ii{\mathrm{i}}

\newcommand{\tend}{\longrightarrow}
\newcommand\dto{\overset{\mathrm{d}}{\tend}}
\newcommand\pto{\overset{\mathrm{p}}{\tend}}
\newcommand\asto{\overset{\mathrm{a.s.}}{\tend}}
\newcommand\eqd{\overset{\mathrm{d}}{=}}
\newcommand\neqd{\overset{\mathrm{d}}{\neq}}
\newcommand\op{o_{\mathrm p}}
\newcommand\Op{O_{\mathrm p}}

\newcommand\bbR{\mathbb R}
\newcommand\bbC{\mathbb C}
\newcommand\bbN{\mathbb N}  %{1,2,...}
\newcommand\bbT{\mathbb T}
\newcommand\bbZ{\mathbb Z}

\newcounter{CC}
\newcommand{\CC}{\stepcounter{CC}\CCx} %new constant C_i
\newcommand{\CCx}{C_{\arabic{CC}}}     %repeats the last C_i
\newcommand{\CCdef}[1]{\xdef#1{\CCx}}     %defines #1 as the last C_i
\newcounter{cc}
\newcommand{\cc}{\stepcounter{cc}\ccx} %new constant c_i
\newcommand{\ccx}{c_{\arabic{cc}}}     %repeats the last c_i
\newcommand{\ccdef}[1]{\xdef#1{\ccx}}     %defines #1 as the last c_i

%\renewcommand\Re{\operatorname{Re}}
%\renewcommand\Im{\operatorname{Im}}

\newcommand\E{\operatorname{\mathbb E{}}}
\renewcommand\P{\operatorname{\mathbb P{}}}
\newcommand\Var{\operatorname{Var}}
\newcommand\Cov{\operatorname{Cov}}
\newcommand\Corr{\operatorname{Corr}}
\newcommand\Exp{\operatorname{Exp}}
\newcommand\Po{\operatorname{Po}}
\newcommand\Bi{\operatorname{Bi}}
\newcommand\Bin{\operatorname{Bin}}
\newcommand\Be{\operatorname{Be}}
\newcommand\Ge{\operatorname{Ge}}
\newcommand\NBi{\operatorname{NegBin}}
%\newcommand\PD{\operatorname{PD}}
\newcommand\Res{\operatorname{Res}}
\newcommand\fall[1]{^{\underline{#1}}}
\newcommand\rise[1]{^{\overline{#1}}}
%\newcommand\Int{\operatorname{Int}}
%\newcommand\arg{\operatorname{arg}}

\newcommand\ga{\alpha}
\newcommand\gb{\beta}
\newcommand\gd{\delta}
\newcommand\gD{\Delta}
\newcommand\gf{\varphi}
\newcommand\gam{\gamma}
\newcommand\gG{\Gamma}
\newcommand\gk{\kappa}
\newcommand\kk{\varkappa}
\newcommand\gl{\lambda}
\newcommand\gL{\Lambda}
\newcommand\go{\omega}
\newcommand\gO{\Omega}
\newcommand\gs{\sigma}
\newcommand\gss{\sigma^2}
%\newcommand\gt{\tau}
\newcommand\gth{\theta}
\newcommand\gthh{\theta^2}
\newcommand\eps{\varepsilon}
\newcommand\ep{\varepsilon}
%\renewcommand\phi{\varphi}

\newcommand\bJ{\bar J}

\newcommand\cA{\mathcal A}
\newcommand\cB{\mathcal B}
\newcommand\cC{\mathcal C}
\newcommand\cD{\mathcal D}
\newcommand\cE{\mathcal E}
\newcommand\cF{\mathcal F}
\newcommand\cG{\mathcal G}
\newcommand\cH{\mathcal H}
\newcommand\cI{\mathcal I}
\newcommand\cJ{\mathcal J}
\newcommand\cK{\mathcal K}
\newcommand\cL{{\mathcal L}}
\newcommand\cM{\mathcal M}
\newcommand\cN{\mathcal N}
\newcommand\cO{\mathcal O}
\newcommand\cP{\mathcal P}
\newcommand\cQ{\mathcal Q}
\newcommand\cR{{\mathcal R}}
\newcommand\cS{{\mathcal S}}
\newcommand\cT{{\mathcal T}}
\newcommand\cU{{\mathcal U}}
\newcommand\cV{\mathcal V}
\newcommand\cW{\mathcal W}
\newcommand\cX{{\mathcal X}}
\newcommand\cY{{\mathcal Y}}
\newcommand\cZ{{\mathcal Z}}

\newcommand\tJ{{\tilde J}}


\newcommand\ett[1]{\boldsymbol1[#1]} 
%\newcommand\ett{\mathbf1} 
\newcommand\etta{\boldsymbol1} 
\def\[#1]{[\![#1]\!]}

\newcommand\smatrixx[1]{\left(\begin{smallmatrix}#1\end{smallmatrix}\right)}
%\newcommand\matrixx[1]{\left(\begin{matrix}#1\end{matrix}\right)}
%\newcommand\matrixx[1]{\begin{pmatrix}#1\end{pmatrix}}


\newcommand\limn{\lim_{n\to\infty}}
\newcommand\limN{\lim_{N\to\infty}}

\newcommand\qq{^{1/2}}
\newcommand\qqq{^{1/3}}
\newcommand\qqw{^{-1/2}}
\newcommand\qqqw{^{-1/3}}
\newcommand\qw{^{-1}}
\newcommand\qww{^{-2}}
%\newcommand\gsi{\sigma^{-1}}

\renewcommand{\=}{:=}

\newcommand\intoi{\int_0^1}
\newcommand\intoo{\int_0^\infty}
\newcommand\intoooo{\int_{-\infty}^\infty}
\newcommand\oi{[0,1]}
\newcommand\ooo{[0,\infty)}
\newcommand\oooo{(-\infty,\infty)}
\newcommand\setoi{\set{0,1}}

\newcommand\dtv{d_{\mathrm{TV}}}

\newcommand\dd{\,\textup{d}}

\newcommand{\pgf}{probability generating function}
\newcommand{\mgf}{moment generating function}
\newcommand{\chf}{characteristic function}
\newcommand{\ui}{uniformly integrable}
\newcommand\rv{random variable}
\newcommand\lhs{left-hand side}
\newcommand\rhs{right-hand side}

\newcommand\gnp{\ensuremath{G(n,p)}}
\newcommand\gnm{\ensuremath{G(n,m)}}
\newcommand\gnd{\ensuremath{G(n,d)}}
\newcommand\gnx[1]{\ensuremath{G(n,#1)}}
\newcommand\citetq[2]{\citeauthor{#2} \cite[{\frenchspacing #1}]{#2}} 

\newcommand\aam{{a_1,\dots,a_m}}
\newcommand\aamp{{a_1+\dots+a_m}}
\newcommand\maam{M_{\aam}}
\newcommand\mab{M_{a,b}}
\newcommand\ax{a^*}
%\newcommand\Ax{{N^*}}
%\newcommand\Ax{{N_*}}
\newcommand\Ax{N_*}
\newcommand\nux{^{(\nu)}}
\newcommand\ab{_{a,b}}
\newcommand\awb{_{a-1,b}}
\newcommand\eith{e^{\ii\gth}}
\newcommand\bfa{\mathbf{a}}
\newcommand\gfd[1]{\gf^{(#1)}}
\newcommand\A{B}
\newcommand\SW{W}
%\newcommand\{}
%\newcommand\{}
%\newcommand\{}
%\newcommand\{}
%QQQ

\newcommand{\Holder}{H\"older}
\newcommand\CS{Cauchy--Schwarz}
\newcommand\CSineq{\CS{} inequality}

\newcommand{\maple}{\texttt{Maple}}

\newcommand\ms[1]{\texttt{[ms #1]}}
%\newcommand{\ms}[1]{\par \vdots \texttt{[ms #1]} \par\vdots}
\newcommand\XXX{XXX \marginal{XXX}}
%\newcommand{\mer}[1]{\texttt{[#1]}\marginal{MER!}}
%\newcommand\REV{\marginal{Revidera!!}}
\newcommand\REM[1]{{\raggedright\texttt{[#1]}\par\marginal{XXX}}}
\newenvironment{OLD}{\Small \REM{Old stuff to be edited:}\par}{}
%\newenvironment{comment}{\setbox0=\vbox\bgroup}{\egroup} %deletes!

\hyphenation{Upp-sala}

%\newcommand\urladdrx[1]{\urladdr{\def~{\~{}}#1}}
%\newcommand\urladdrx[1]{{\urladdr{\def~{{\tiny$\sim$}}#1}}}
\newcommand\urladdrx[1]{{\urladdr{\def~{\Tilde}#1}}}


%begin macros
%\def\L{{\cal L}}
%\baselineskip=14pt
%\parskip=10pt
%\def\halmos{\hbox{\vrule height0.15cm width0.01cm\vbox{\hrule height
%  0.01cm width0.2cm \vskip0.15cm \hrule height 0.01cm width0.2cm}\vrule
%  height0.15cm width 0.01cm}}
%\font\eightrm=cmr8 \font\sixrm=cmr6
%\font\eighttt=cmtt8
%\magnification=\magstephalf
%\def\N{{\cal N}}
%\def\P{{\cal P}}
%\def\Q{{\cal Q}}
%\def\R{{\cal R}}
%\def\1{{\overline{1}}}
%\def\2{{\overline{2}}}
%\renewcommand\Tilde{\char126\relax}
\renewcommand\Tilde{{\tiny$\sim$}}

\title{On the Statistics of the Number of Fixed-Dimensional Subcubes in a Random Subset of the n-Dimensional Discrete Unit Cube}
\headlinetitle{Statistics of the Number of Subcubes in Subsets of the n-Cube}


\date{Feb. 17, 2023} % (typeset \today{} \klockan)} %; revised ...


\authorone{Svante Janson}
\addressone{Department of Mathematics, Uppsala University, PO Box 480,
SE-751~06 Uppsala}
\countryone{Sweden}
\emailone{svante.janson@math.uu.se}
%\urladdrx{\href{http://www.math.uu.se/~svante/}{http://www.math.uu.se/~svante/}}


\authortwo{Blair Seidler}
\addresstwo{ Department of Mathematics,
Rutgers University (New Brunswick), Piscataway NJ 08854}
\countrytwo{U.S.A.}
\emailtwo{blair@math.rutgers.edu}

\authorthree{Doron Zeilberger}
\addressthree{Department of Mathematics, Rutgers University
(New Brunswick), Piscataway, NJ 08854}
\countrythree{U.S.A.}
\emailthree{zeilberg@math.rutgers.edu}



%\keywords{<keywords>}
%\subjclass[2000]{} 
%{60C05 (68P10,68W40)} %%{Primary: <subject>; Secondary: <subject>}

\begin{comment}  % Some suggestions:
05 Combinatorics 
05C Graph theory [For applications of graphs, see 68R10, 90C35, 94C15]
05C05 Trees
05C07 Degree sequences
05C35 Extremal problems [See also 90C35]
05C40 Connectivity
05C65 Hypergraphs
05C80 Random graphs
05C90 Applications

60 Probability theory and stochastic processes
60C Combinatorial probability
60C05 Combinatorial probability

60F Limit theorems [See also 28Dxx, 60B12]
60F05 Central limit and other weak theorems
60F17 Functional limit theorems; invariance principles

60G Stochastic processes
60G09 Exchangeability
60G55 Point processes

60J Markov processes
60J65 Brownian motion [See also 58J65]
60J80 Branching processes (Galton-Watson, birth-and-death, etc.)

60K Special processes
60K35 Interacting random processes; statistical mechanics type models;
      percolation theory [See also 82B43, 82C43] 

68 Computer science
68P Theory of data
68P05 Data structures
68P10 Searching and sorting

68Q Theory of computing
68Q25 Analysis of algorithms and problem complexity [See also 68W40] 

68R Discrete mathematics in relation to computer science
68R10 Graph theory [See also 05Cxx, 90B10, 90B35, 90C35]

68W Algorithms [For numerical algorithms, see 65-XX; for combinatorics and graph theory, see 68Rxx]
68W40 Analysis of algorithms [See also 68Q25]

82 Statistical mechanics, structure of matter
82B Equilibrium statistical mechanics
82B10 Quantum equilibrium statistical mechanics (general)
82B26 Phase transitions (general)

92 Biology and other natural sciences
92D Genetics and population dynamics
92D30 Epidemiology
\end{comment}


\begin{document}
\begin{thebibliography}{99}

%\bibitem[Andrews(1976)]{Andrews} 
%G. E. Andrews, 
%{\it The Theory of Partitions}, Addison-Wesley, 
%Reading, Mass., 1976.

%\bibitem{FellerI}
%W. Feller, 
%\emph{An Introduction to Probability Theory and Its Application}, volume I, 
%third edition, Wiley, %John Wiley \& Sons, 
%New York, 1968.

\bibitem{SJ58}
Svante Janson, 
Normal convergence by higher semiinvariants with applications to sums of
dependent random variables and random graphs. 
\emph{Annals of Probability} \textbf{16} (1988), 305--312.

\bibitem{JLR}
Svante Janson, Tomasz {\L}uczak and Andrzej Ruci{\'n}ski,
\emph{Random Graphs}.
Wiley, New York, 2000. 
%ISBN: 0-471-17541-2.

\bibitem{UK}
Urszula Konieczna,
Asymptotic normality of subcubes in random subgraphs of the $n$-cube.
\emph{Discrete Mathematics} \textbf{121} (1993), 117--122.

\bibitem{LeonovShiryaev}
V. P. Leonov and Albert Nikolaevich Shiryaev,
%Leonov, V. P.; Sirjaev, A. N.
On a method of semi-invariants.
\emph{Teor. Veroyatnost. i Primenen} \textbf4 (1959), 342--355.
(Russian.)
%Sur le calcul des semi-invariants. (Russian. French summary)
%MR0114234 
English transl:
%On a method of semi-invariants.
\emph{Theor. Probability Appl.} \textbf4 (1959), 319--329.
%MR0123345

\bibitem{T}
Thotsaporn Aek Thanatipanonda, {\it Reviews of Symbolic Moment Calculus},    arXiv:2003.11749.
%\href{https://arxiv.org/abs/2003.11749}{\tt https://arxiv.org/abs/2003.11749}

\bibitem{KW}
Karl Weber, 
Poisson convergence in the $n$-cube. 
\emph{Math. Nachr.} \textbf{131} (1987), 49--57.
%MR0908798



\end{thebibliography}

\centerline{Communicated  by  Ayman Badawi}
\vskip0.2in

\footnotesize MSC 2010 Classifications: Primary 05A15; Secondary 05A16, 68R10.

\vskip0.1in

Keywords: Boolean function, discrete unit cube, subcube, expectation, variance,
asymptotic normality, linearity of expectation, experimental mathematics, enumeration,
central moment, cumulant

\vskip0.1in

\normalsize

 {\bf Abstract} This paper consists of two independent, but related parts.
In the first part we show how to use symbolic computation to derive explicit expressions for the first few moments
of the number of implicants that a random Boolean function has, or equivalently
the number of fixed-dimensional subcubes contained in a random subset of the
$n$-dimensional cube. These explicit expressions suggest, but do not prove,
that these random variables are always asymptotically normal.

The second part presents a full, human-generated proof, of this asymptotic
normality, first proved by Urszula Konieczna.

Accompanied by Maple package SMCboole.txt, available from
\hfill\break
\href{http://www.math.rutgers.edu/~zeilberg/mamarim/mamarimhtml/subcubes.html}{\texttt{
http://www.math.rutgers.edu/\Tilde zeilberg/mamarim/mamarimhtml/subcubes.html}}
\vspace*{5mm}

%\dedicatory{}


\maketitle


\section{Introduction}\label{S:intro}


\subsection{Motivation}

Recall that an \emph{implicant} of a Boolean function in $n$ variables, $f(x_1, \dots, x_n)$ is a pure disjunction 
\begin{align}
x_{i_1}^{a_1} \,\wedge \,x_{i_2}^{a_{2}} \wedge \dots \wedge x_{i_r}^{a_r} ,
\end{align}
that implies it. 
Here $1 \leq i_1 < \dots i_r \leq n$, $a_1, \dots, a_r \in \{0,1\}$, $z^1=z$, and $z^0=\bar{z}$ (the negation of $z$).

Fix $r$ and let $n$ vary. We are interested in the statistical distribution of the random variable {\it number of implicants of length $n-r$} in a uniformly-at-random Boolean function
of $n$ variables. Clearly, when $r=0$ it is nothing but the good old (fair) binomial distribution with $2^n$ {\it fair coin-tosses}, $B(\frac{1}{2},2^n)$.

Equivalently, for a random subset of the $n$-dimensional cube, we are interested in the statistical distribution of the number of $r$-dimensional subcubes properly contained in it.

We would like to have \textbf{explicit expressions}, in $n$, for the $k^{th}$ moment of this random variable, for as many as possible $r$ and $k$. This turns out to be
a challenging {\it symbolic-computational} problem that we will address in the first part of this paper.

In the second part we consider asymptotics as \ntoo.
It was proved by  Urszula Konieczna \cite{UK} that 
for each fixed $r$, this distribution is {\it asymptotically normal}.
We will reprove this by a somewhat different method which leads to a simple
proof.

\subsection{Our Random Variables}

The \emph{sample space} is the set of subsets of $\{0,1\}^n$, that has cardinality $2^{2^n}$.

Let's define our random variables formally.
For a (uniformly-at-) random subset $S$, of $\{0,1\}^n$, and \emph{fixed} $r$, define the \textbf{random variable}
\begin{align}
X_r(S):=\text{ number of }r\text{-dimensional subcubes of }S.
\end{align}

For example, if $n=3$ and
\begin{align}
S=\{ 000,001,010,011,100,111\},
\end{align}
we have
\begin{align}
X_0(S)=6 , \quad
X_1(S)=6 , \quad
X_2(S)=1 , \quad
X_3(S)=0 .
\end{align}


We would like to get, for as many pairs $(k,r)$ as possible, {\bf explicit} expressions in $n$, for the $k$-th moment of $X_r$, i.e. for
\begin{align}
f_{kr}(n):=\E[{X_r}^k](n) .
\end{align}


\section{Explicit Expressions for Moments of the Number of Low-Dimensional Subcubes in a Random Subest of the $n$-dimensional cube}\label{S:Expl}


\subsection{The Expectation and Variance}

The {\it first} moment, aka  {\it expectation}, aka {\it mean}, aka {\it average}, is easy, using {\bf linearity of expectation}.

For any specific subcube $C$ of $\{0,1\}^n$, define the {\it atomic} random variable, $X_C$, on subsets, $S$, of $\{0,1\}^n$ as follows.
\begin{align}
X_C(S)\,= \,
\begin{cases}
1,&\text{if} \quad  C \subset S ; \\
0,& \text{otherwise}.
\end{cases}
\end{align}
Let $\cC(n,r)$ be the set of all ${\binom{n}{r}}2^{n-r}$ $r$-dimensional subcubes of $\{0,1\}^n$. Then, of course,
\begin{align}\label{xr}
X_r(S) \, = \, \sum_{C \in \cC(n,r)} X_C(S) .  
\end{align}
Applying the {\it expectation} functional and using the {\bf linearity of expectation}, we get that the average, let's call it 
$\mu_r(n)$, is
\begin{align} \label{LoA}
\mu_r(n) =
\E[X_r] \, = \, \E\lrsqpar{\sum_{C \in \cC(n,r)} X_C} \, = \, \sum_{C \in \cC(n,r)} \E[X_C] .
\end{align}

Now the probability that a random subset of $\{0,1\}^n$ contains an $r$-dimensional subcube $C$ is 
$\left(\frac{1}{2}\right)^{2^r}$, since for each of its vertices, the chance of it belonging to $S$ is $\frac{1}{2}$,
and by {\it independence} the probability that all its $2^r$ vertices belong to $S$ is indeed  $\frac{1}{2^{2^r}}$. The probability that $X_C(S)=0$ is of course   $1-\left(\frac{1}{2}\right)^{2^r}$, hence
\begin{align}\label{mu}  
\E[X_C]= 1 \cdot \left(\frac{1}{2}\right)^{2^r} \, + \, 0 \cdot \left(1-\left(\frac{1}{2}\right)^{2^r}\right) 
\, = \, \left(\frac{1}{2}\right)^{2^r} .
\end{align}
Going back to equation \eqref{LoA} we have
\begin{align}
\mu_r(n) \, = \, \sum_{C \in \cC(n,r)} \E[X_C] = \, \sum_{C \in \cC(n,r)} \frac{1}{2^{2^r}} = |\cC(n,r)| \cdot  \frac{1}{2^{2^r}} = \frac{{\binom{n}{r}}2^{n-r}}{2^{2^r}} .
\end{align}


In a beautiful paper, Thanatipanonda \cite{T} derived an explicit expression for the general second moment, for every $r$-dimensional cube.\\\\

{\bf Thanatipanoda's General Formula for the Second Moment}:
\begin{align}\label{th2}
\E[X_r^2] \, = \,
\sum_{i=0}^{r} \frac{n! 2^{n-i}}{i!(r-i)!^2 (n-2r+i)! 2^{2^{r+1}}} \cdot (2^{2^i} -1) \, 
+ \,\frac{[{\binom{n}{r}} 2^{n-r}]^2}{2^{2^{r+1}}} ,
\end{align}
from which immediately follows, using $[\E(X_r-\mu_r(n))^2]=\E[X_r^2]-\mu_r(n)^2$, the following formula.\\\\

{\bf Thanatipanoda's General Formula for the Variance}:
\begin{align}\label{thvar}
  \text{Var}(X_r)=
\sum_{i=0}^{r} \frac{n! 2^{n-i}}{i!(r-i)!^2 (n-2r+i)! 2^{2^{r+1}}} \cdot
  (2^{2^i} -1)  .
\end{align}
Note that the variance is a polynomial in $(n,2^n)$ of degree $2r$ in $n$ and degree $1$ in $2^n$.

\subsection{Higher Moments}

\subsubsection{Edges}

Thanatipanonda was unable to get such a general formula for higher moments, but did get $\E[X_1^3]$, from which he immediately deduced that
the third central moment (or third-moment-about-the-mean) of $X_1$ is
\begin{align}
\E[(X_1-\mu_1(n))^3] \, = \, \frac{3n^3 2^n}{64} .
\end{align}


Using the symbolic-computational algorithms to be described in the next section, we managed to derive the following explicit formulas
\begin{align}
\E[(X_1-\mu_1(n))^4] =\frac{n 2^{n}}{1024}\big(&12 n^{3} 2^{n} +12 n^{2} 2^{n}+40 n^{3} \notag\\
&+3 n 2^{n}-48 n^{2}+12 n -16\big),
\end{align}

\begin{align}
\E[(X_1-\mu_1(n))^5]  = \frac{5 n^{3} 2^{n} }{1024} \left(6 n^{2}2^{n} +3 n 2^{n}+4 n^{2}-24 n +8\right),
\end{align}

\begin{align}
\E[(X_1-&\mu_1(n))^6] = \notag\\
\frac{n 2^{n}}{32768}\cdot\big(&120 n^{5}\left(2^{n}\right)^{2} +180 n^{4} \left(2^{n}\right)^{2}
+1920 n^{5} 2^{n}+90 n^{3}\left(2^{n}\right)^{2} \notag\\
&-840 n^{4} 2^{n}-1792 n^{5}
+15 n^{2} \left(2^{n}\right)^{2}-360 n^{3} 2^{n} -5280 n^{4}\notag\\
&-300 n^{2} 2^{n}+3840 n^{3}
-240 n 2^{n}+3840 n^{2}-6720 n +4864 \big)
.\end{align}
It follows that the {\bf scaled moments about the mean} for the third, fourth, fifth, and sixth moments, converge, as $n \rightarrow \infty$, to $0,3,0,15$ respectively,  the  respective moments of the normal distribution,
indicating that the random variable $X_1$ (the number of edges contained in $S$)  is {\bf probably} asymptotically normal. To fully prove asymptotic normality, of course, we need to prove
it for all moments, not just for the first six.


\subsubsection{Squares}

We only managed to get explicit expressions for the third and fourth moments for $X_2$.
\begin{align}
\E[(X_2-\mu_2(n))^3] \, = \, 
\frac{2^{n} n \left(n -1\right)}{32768} \left(9 n^{4}+6 n^{3}+21 n^{2}-16 n -34\right),
\end{align}

\begin{align}
\E[(X_2-\mu_2(n))^4&] = \notag\\
\frac{2^{n} n \left(n -1\right)}{4194304} \cdot
\big(&12 n^{6} 2^{n}+12 n^{5} 2^{n} +
520 n^{6}+24 n^{4} 2^{n}-24 n^{5}-12  n^{3}2^{n}+1272 n^{4}
\notag\\
&-9 n^{2}2^{n} -840 n^{3}-27 n 2^{n}-5232 n^{2}-2768 n +240\big)
.\end{align}


\subsubsection{3-dimensional cubes}

We only managed to get an explicit expression for the third moment for $X_3$.
\begin{align}
\E[(X_3-\mu_3(n))^3] =\quad\quad& \notag\\
\frac{2^{n} n \left(n -1\right) \left(n -2\right)}{2415919104} \big(&14 n^{6}+24 n^{5}+479 n^{4}+2046 n^{3}\notag\\
&+6779 n^{2}+15444 n -23112\big)
.\end{align}


\subsection{Our Method}

Obviously we did not derive these formulas by hand. We had to teach our computer how to find them. It also uses {\it linearity of expectation}, but
with higher moments things get very complicated.
Recall that
\begin{align}
X_r(S) \, = \, \sum_{C \in \cC(n,r)} X_C(S) .
\end{align}
Hence, the $k$-th moment is
\begin{align}\label{b1}
  \E\left[(X_r)^k\right]&=
\E\Big[\Big( \sum_{C \in \cC(n,r)} X_C(S) \Big)^k\Big]
\notag\\&
= \sum_{[C_1, \dots, C_k] \in \cC(n,r)^k} \E[X_{C_1} X_{C_2} \cdots X_{C_k}]
.\end{align}

So we sum over all $({\binom{n}{r}}2^{n-r})^k$ members of $\cC(n,r)^k$.  
Since the product $X_{C_1}(S) X_{C_2}(S) \cdots X_{C_k}(S)=1$ if {\bf each} of $C_1,C_2, \dots, C_k$ is properly included in $S$ (i.e. if
each vertex  in $C_1 \cup C_2 \cup \dots C_k$ belongs to $S$), and $0$ otherwise, 
the contribution,
or weight,
due to each such term is
\begin{align}
\text{Weight}([C_1,C_2,\dots,C_k])=
\E\left[X_{C_1} X_{C_2} \cdots X_{C_k}\right]= \frac{1}{2^{|C_1 \cup C_2 \cup \dots C_k|}} .
\end{align}


\subsubsection{Data Structure}

Every $r$-dimensional subcube of $\{0,1\}^n$ has the form
\begin{align}
C=\{ (x_1, \dots, x_n) \in \{0,1\}^n \, | \,x_{i_1}=\alpha_{i_1}, \dots, x_{i_{n-r}}=\alpha_{i_{n-r}} \} ,
\end{align}
for some $1\leq i_1 < i_2 < \dots <i_{n-r} \leq n$ and $(\alpha_{i_1}, \dots \alpha_{i_{n-r}}) \in \{0,1\}^{n-r}$.
A good way to represent it on a computer is as a row-vector of length $n$, in the {\it alphabet} $\{0,1,*\}$,
where the entries corresponding to  $i_1, i_2, \dots, i_{n-r}$  have $\alpha_{i_1}, \dots. \alpha_{i_{n-r}}$ respectively
and the remaining $r$ entries are filled with {\bf wild cards}, denoted by $*$.

For example, if $n=7$ and $r=3$, the $3$-dimensional cube
\begin{align}
\{ (x_1, \dots, x_7) \in \{0,1\}^7 \, | \,x_2=1, x_4=1, x_5=0, x_7=1\} ,
\end{align}
is represented by
\begin{align}
*1*10*1 .
\end{align}

We are trying to find a weighted count of {\bf ordered} $k$-tuples of $r$-dimensional subcubes. The natural data structure for these is the set of $k$ by $n$ matrices in the
`alphabet' $\{0,1,*\}$ where every row has exactly $r$ `wildcards', *.

Let's call this set of matrices, that correspond to $\cC(n,r)^k$, $\cC(n,r,k)$.

For any {\bf specific}, numeric $n$, there are `only'  $(2^{n-r}{\binom{n}{r}})^k$ of these matrices, and for each and every one of them one can find
the cardinality of the union of the corresponding subcubes, let's call it $v$, and add to the running sum $\frac{1}{2^v}$. But we want to do it for
{\bf symbolic} $n$, i.e. for `all' $n$. 
We will soon see how, for each {\it specific} (numeric) $r$ and $k$ this can be done {\it in principle}, but only for relatively small $r$ and $k$ {\it in practice}.
But let's try and push it as far we can.
An interesting consequence of our algorithm is the precise degree in $n$ and $2^n$ of the expression for $\E[X_r^k](n)$.


\subsubsection{The Kernel}


A key object in our approach is the {\bf kernel}. 
Given a $k \times n$ matrix $M$ in the alphabet $\{0,1,*\}$ let's call a column {\bf active} if it contains
at least one `$*$'. Note that the matrix has exactly $k\cdot r$  `$*$'s, hence the number of {\bf active columns}, let's call it $a$,
is between $r$ and $k\cdot r$. \footnote{More generally, if we want to find an expression for the {\it mixed moment} $\E[X_{r_1} \cdots X_{r_k}]$
the number of active columns is between $\max(r_1, \dots, r_k)$ and $r_1 + \dots + r_k$.} The {\bf kernel} of $M$ is
the submatrix of $M$ consisting of its active columns.\\

Let $\cC_a(n,r,k)$ be the subset of $\cC(n,r,k)$ matrices with exactly $a$ active columns. We will call such a matrix in {\bf canonical form} if
the active columns are occupied by the $a$ leftmost columns (i.e. its kernel is contiguous starting in the first column).
Let's denote by $\overline{\cC}_a(n,r,k)$ the set of such matrices in canonical form.
Obviously, there are ${\binom{n}{a}}$ ways to choose which of the $n$ columns are active and hence
\begin{align}
\text{Weight}(\cC(n,r,k)) \, =& \, \sum_{a=r}^{rk} \text{Weight}(\cC_a(n,r,k)) \notag\\
=&\sum_{a=r}^{rk} {\binom{n}{a}} \, \text{Weight}(\overline{\cC}_a(n,r,k)) .
\end{align}
For any set, $S$, $\text{Weight}(S)$ is the sum of the weights of its members. Note that this has degree $rk$ in $n$.

It remains to do a {\it weighted-count}, where every matrix gets `credit' $1/2^v$, where $v$ is the cardinality of the union of the corresponding subcubes represented by the $k$ rows,
for the set $\overline{\cC}_a(n,r,k)$, of matrices in canonical form. Note that there are only {\bf finitely many} choices for the $a$ leftmost columns, i.e. the set of
$k \times a$ matrices in the alphabet $\{0,1,*\}$ with the property that every column has at least one `$*$', and every row has exactly $r$ `$*$'s.
These can be divided into {\it equivalence classes} obtained by permuting rows and columns and transposing $0$ and $1$ in any given column. Once these are sorted into equivalence classes, one needs only
examine one representative, and then multiply the weight by the cardinality of the class.

But what about the $n-a$ rightmost columns? There are $2^{k(n-a)}$ possible submatrices; the alphabet here is $\{0,1\}$.
Almost all of these have distinct rows, more precisely,
\begin{align}
{\binom{2^{n-a}}{k}} k!
\end{align}
of them, and these will produce the smallest possible weight in conjunction with any kernel. The other extreme is that all the rows of the submatrix consisting of the $n-a$ rightmost columns are identical, and then there are only $2^{n-a}$ choices to fill them in.

In general, every such member of $\overline{\cC}_a(n,r,k)$, determines a {\bf set partition} of the set of rows $\{1,\dots,k\}$, where two rows are {\it roomates} if they have the same last $n-a$ entries.
If that set-partition has $m$ members
$1 \leq m \leq k$, then the number of choices of assigning {\bf different} $0-1$ vectors of length $n-a$ to each of the parts of the set-partition is
\begin{align}
{\binom{2^{n-a}}{m}} m! .
\end{align}

Now for each $a$ and for each set-partition, we let the computer generate the {\bf finite} set of $k \times a$ matrices in the alphabet $\{0,1,*\}$. Each of the
members of the set partition has its own submatrix, and we ask our computer to kindly find the number of vertices in the corresponding union of
subcubes corresponding to each member of the examined set partition. Since they are disjoint, we add them up, getting $v$ for that particular pair (matrix, set-partition), giving credit $1/2^v$.

\subsubsection{Implementation}

All this is implemented in the Maple package {\tt SMCboole.txt}, available from:
\\
\href{https://sites.math.rutgers.edu/\~zeilberg/tokhniot/SMCboole.txt}{\tt https://sites.math.rutgers.edu/\~{}zeilberg/tokhniot/SMCboole.txt}

In particular `{\tt Moms(A,n);}', for any list of non-negative integers $A=[r_1, \dots r_k]$ gives you the mixed moment $\E[X_{r_1} \cdots X_{r_k}]$. For example, to get the
third moment of the number of edges (i.e. $1$-dimensional subcubes) type \quad
{\tt Moms([1,1,1],n);}  \quad, which very quickly returns:
\begin{align}
\frac{2^{n} n^{2} \left(\left(2^{n}\right)^{2} n +12 \,2^{n} n +6 \,2^{n}+24 n \right)}{512} .
\end{align}

To get the third moment of the number of squares (i.e. $2$-dimensional subcubes), type \quad {\tt Moms([2,2,2],n);} \quad,
getting
\begin{align}
\frac{2^{n} n  \left(n -1\right)}{2097152} \cdot\bigg(&\left( 2^{n}\right)^{2} n^{4}-2 \left(2^{n}\right)^{2} n^{3}
+48 \,2^{n} n^{4}+\left(2^{n}\right)^{2} n^{2} \notag\\
&+576 n^{4}+24 \,2^{n} n^{2}+384 n^{3}-72 \,2^{n} n \notag\\
&+1344 n^{2}-1024 n -2176\bigg) .
\end{align}

The third moment of the number of $3$-dimensional cubes takes a bit longer, and we were unable to compute the fourth moment of the number of $3$-dimensional cubes; it took too much time and too much space.

More informative for {\it statistical purposes} are the {\bf central moments}, $\E[(X_r-\mu_r(n))^k]$, that  Maple easily derives,
using {\it linearity of expectation} from the pure moments. The function call for this is
{\tt MOMrk(r,k,n);} where {\tt r} and {\tt k} are {\it numeric} but $n$ is a {\bf symbol} denoting the dimension of the ambient cube.

To get the explicit expression given above 
for the third through sixth moments for the number of edges, the third and fourth moments for the number of squares, and the third moment for the number of $3$-dimensional
subcube (all {\it about the mean}) we typed:

{\tt MOMrk(1,3,n);} ,  {\tt MOMrk(1,4,n);} ,  {\tt MOMrk(1,5,n);} ,  {\tt MOMrk(1,6,n);} , 

{\tt MOMrk(2,3,n);} ,  {\tt MOMrk(2,4,n);} ,  

{\tt MOMrk(3,3,n);} 

respectively. To our chagrin, `{\tt MOMrk(3,4,n);}'   took too long.


\subsubsection{Consequence of the algorithm}: The $k$-th  moment of $X_r$ is a bivariate polynomial in $(n,2^n)$ of degree $k\,r$ in $n$ and degree $k$ in $2^n$ .

This raises the {\it theoretical} possibility (in God's computer) of finding these expressions by {\bf pure brute force}. The generic polynomial in $(n,2^n)$ of degree $kr$ in $n$ and degree $k$ in $2^n$
has $(1+kr)(1+k)$ `degrees of freedom'. So using {\it undetermined coefficients} we need to generate a table of $\E[X_r^k](n)$ for $1 \leq n \leq (1+kr) \cdot (1+k)$. 
After gathering the data, we use linear algebra to solve a system of  $(1+kr) \cdot (1+k)$ equations with that many unknowns.
For each specific
$n=n_1$ there are `only' $2^{2^{n_1}}$ subsets, and for each of them we can ask how many $r$-dimensional subcubes do they contain, raise it to the $k$-th power and take the average.
Alas, $2^{2^5}$ is already big enough, so only God's computer, with practically infinite time and space,  can carry this brute force approach.





\section{Asymptotic Normality }\label{S:ANor}

As said in the introduction,
asymptotic normality of $X_{r}$ was proved by  Urszula Konieczna \cite{UK}. 
The random variable $X_r$ depends on $n$, and for emphasis, we denote it in
the sequel by $X_{n,r}$.
We may then state the result as follows,
where $N(0,1)$ denotes
the standard normal distribution.


\begin{theorem}[\cite{UK}]\label{T1}
For every fixed $r\ge0$, %the normalized random variables 
we have as \ntoo
\begin{align}\label{t1}
\frac{X_{n,r}-\mu_r(n)}{\sqrt{\Var\xpar{X_{n,r}}}}
\dto N(0,1)  
\end{align}
with convergence in distribution
and of all moments.  
\end{theorem}

As already noted at the very beginning of this paper, the case $r=0$ is
well known, since then the variables $X_C$ are 
independent and this becomes an instance of the central limit theorem for
binomial variables, known since a pamphlet by de Moivre in 1733.
(This case can be used as a sanity check in the arguments below.)


The proof in \cite{UK} is based on the method of moments.
We will in this section survey how a variation of this method, based on
{\bf cumulants} and {\bf dependency graphs} leads to a quick proof with
almost no calculations.

\begin{remark}
Urszula Konieczna \cite{UK} 
considered a more general case where each subcube appears with
probability $p\in(0,1)$, where $p=p(n)$ may depend on $n$, and showed
asymptotic normality for a large range of $p(n)$.
A special case, not including the case $p=\frac12$ treated here, was earlier
shown by Karl Weber \cite{KW}.
This extension can also be treated by the method below, but for simplicity
we continue to consider only the case $p=\frac12$ discussed in \refS{S:Expl}.
\end{remark}


The aim is thus to show
that the leading terms in the expressions for the
central moments 
$\E[(X_{n,r}-\mu_r(n))^k]$ are what we would expect from a normal distribution, or,
more formally, that the scaled central moments
\begin{align}\label{a1}
\E\lrsqpar{\lrpar{\frac{X_{n,r}-\mu_r(n)}{\sqrt{\Var\xpar{X_{n,r}}}}}^k}
\end{align}
converge as \ntoo{}
to the corresponding moment $\E [Z^k]$ of a standard normal random
variable $Z$, for every fixed $r\ge0$ and $k\ge1$.
Then, the random variables $X_{n,r}$ are asymptotically normal by the method of
moments.


We thus want to find
the leading term of the moment \eqref{a1}. In \refS{S:Expl}, we computed the
central moments from the pure moments. This involves massive cancellation of
high-order terms, and is less suitable for a (human) proof for general $k$.

As a first step, we may modify \eqref{b1} and instead expand the central
moment as
\begin{align}\label{b1b}
  \E\bigsqpar{(X_{n,r}-\mu_r(n))^k}&
%=\E\left[\left( \sum_{C \in \cC(n,r)} X_C(S) \right)^k\right]
%\notag\\&
= \sum_{[C_1, \dots, C_k] \in \cC(n,r)^k} \E\bigsqpar{(X_{C_1}-\mu) \cdots (X_{C_k}-\mu)}
,\end{align}
where $\mu=\E[X_C]$ is given by \eqref{mu}.
Note that two variables $X_{C_1}$ and $X_{C_2}$ are independent if
$C_1\cap C_2=\emptyset$. In particular, for $k=2$, all such terms in the sum
in \eqref{b1b} vanish; this leaves only 
$\Theta(n^{2r}2^n)$ terms of the $\Theta\bigpar{n^{2r}2^{2n}}$ terms in the
full sum, or in \eqref{b1}, which is reflected in the leading terms in
\eqref{th2} and \eqref{thvar}.

For higher moments, we can obtain a further reduction by considering 
{\bf cumulants} (also called {\bf semiinvariants}) instead of moments.
We describe this briefly, referring to e.g.\
\cite{LeonovShiryaev} or \cite[pp.~145--149]{JLR} for details.
In general, if $Y$ is a  random variable with moments $m_k:=\E [Y^k]$ and
moment generating function $\sum_{k=0}^\infty m_k t^k/k!$, then the cumulants
$\kk_k=\kk_k(Y)$ are the coefficients of the generating function
\begin{align}\label{kk}
  \sum_{k=1}^\infty\kk_k \frac{t^k}{k!}
:=
\log \lrpar{  \sum_{k=0}^\infty m_k \frac{t^k}{k!}}.
\end{align}
(The generating functions can be regarded as  formal power series.)
Moreover, this generalizes to mixed cumulants $\kk_k(Y_1,\dots,Y_k)$ of
several random variables.
This means that there are algebraic relations: each cumulant
$\kk_k$ is a polynomial
in the moments $m_\ell$ of order $\ell\le k$, and conversely, and this
generalizes to mixed cumulants and moments.
For example, $\kk_1=m_1=\E Y$, and $\kk_2=m_2-m_1^2=\Var(Y)$, and the mixed
cumulant $\kk(Y_1,Y_2)=\E[Y_1Y_2]-\E[Y_1]\E[Y_2]$, the covariance of $Y_1$
and $Y_2$.

A convenient property of cumulants is that a normal distribution
$N(\mu,\gss)$ has moment generating function 
$\exp\bigpar{\mu t + \tfrac{\gss}2t^2}$, and thus by \eqref{kk} cumulants
$\kk_k=0$ for $k\ge3$. Consequently, to show that a normalized sequence of
random variables is asymptotically normal, it suffice to show that each
cumulant $\kk_k$ with $k\ge3$ converges to 0.

The mixed cumulants are multilinear, and thus we have an analog of
\eqref{b1} and \eqref{b1b}:
\begin{align}\label{kkk}
\kk_k\bigpar{X_{n,r}}&
=\kk\bigpar{X_{n,r},\dots,X_{n,r}}&
= \sum_{[C_1, \dots, C_k] \in \cC(n,r)^k} \kk\bigpar{X_{C_1}, \dots, X_{C_k}}
.\end{align}
We can here reduce the number of terms in the sum by a general
property of cumulants: If there is a partition $\set{1,\dots,k}=I\cup J$
into two non-empty sets $I$ and $J$ such the two families of random
variables
$(Y_i\mid i\in I)$ and $(Y_j\mid j\in J)$ are independent of each other,
then the mixed cumulant $\kk(Y_1,\dots,Y_k)=0$.
(Informally, this means that the $k$th cumulant $\kk_k(X_{n,r})$ can be seen
as some kind of measure of $k$th order dependencies in the sum \eqref{xr},
with lower order dependencies removed.)

As noticed above, each of our variables $X_C$ is independent of most of the
others. We need to keep track of not just pairwise independence, but also
independence between families of such variables; it is then convenient to
use the notion of {\bf dependency graphs}.

\begin{definition}
  Let $(Y_\ga)_{\ga\in\cA}$ be a family of random variables, with some
  arbitrary index set $\cA$.
A  {\bf dependency graph} for this family is a graph $\gG$ with vertex set
$\cA$, such that if $I$ and $J$ are two disjoint subsets of $\cA$ such that
there is no edge in $\gG$ with one endpoint in $I$ and the other in $J$,
then the 
two families $(Y_\ga\mid \ga\in I)$ and $(Y_\gb\mid \gb\in J)$ 
are independent of each other.
\end{definition}


\begin{lemma}\label{LgG}
  Let $\gG(n,r)$ be the graph with vertex set $\cC(n,r)$ and an edge between
  two distinct cubes $C,C'\in\cC(n,r)$ iff\/ $C\cap C'\neq\emptyset$.
Then $\gG(n,r)$ is a dependency graph for the family $\xpar{X_C\mid C\in\cC(n,r)}$.
\end{lemma}
\begin{proof}
Let $I$ and $J$ be disjoint subsets of $\cC(n,r)$ with no edge between $I$
and $J$.
Then $V_I:=\bigcup_{C\in I}C$  and $V_J:=\bigcup_{C\in J}C$ are two disjoint
subsets of $\setoi^n$. Thus the two random sets $S\cap V_I$ and $S\cap V_J$
are independent. Since the random variables $\xpar{X_C\mid C\in I}$ only
depend on $S\cap V_I$, and similarly for $J$, 
it follows that the families
$\xpar{X_C\mid C\in I}$ and $\xpar{X_C\mid C\in J}$ are independent.
\end{proof}

We can now use the dependency graphs $\gG(n,r)$ and the machinery above to
estimate cumulants of $X_{n,r}$ and show that
$\kk_k\bigpar{X_{n,r}}=o\bigpar{(\Var X_{n,r})^{k/2}}$ as \ntoo{} for every
fixed $r\ge0$ and $k\ge3$. However, we do not have to do these calculations,
since they already have been done under general hypotheses, leading to
convergence theorems that are easy to apply in situations like ours where
there is a rather sparse dependency graph.
We use here the following theorem, 
taken (with minor changes in notation) from %\cite{SJ58}
\cite[Theorem 2, with moment convergence by its proof and Theorem 1]{SJ58}.
See \eg{} \cite[Theorems 6.18 and 6.20]{JLR} for some related theorems 
(proved by the same cumulant method) that also can be used to show \refT{T1},
and \cite[Theorem~6.33]{JLR} for another related theorem (proved by a
different method) that also yields the asymptotic normality in \refT{T1}
(but not immediately moment convergence).

\begin{theorem}[\cite{SJ58}]\label{T58}
  Suppose that, for each $n$,
$\xpar{Y_{n,\ga}\mid \ga\in\cA_n}$ is a family of bounded random variables:
$|Y_{n,\ga}|\le A_n$. Let $N_n:=|\cA_n|$.
Suppose further that $\gG_n$ is a dependency graph for this family, and
let $M_n$ be the maximal degree of $\gG_n$ (assuming this is not $0$, in which
case we let $M_n:=1$). Let $\SW_n=\sum_{i\in\cA_i} Y_{n,i}$ and
$\gss_n:=\Var(\SW_n)$.
If there exists a positive integer $m$ such that
\begin{align}
  \label{t58a}
(N_n/M_n)^{1/m} M_n A_n/\gs_n \to0
\qquad\text{as \ntoo},
\end{align}
then
\begin{align}
\bigpar{\SW_n-\E \SW_n}/\gs_n\dto N(0,1)
  \qquad\text{as \ntoo}
,\end{align}
with convergence in distribution and with all moments.
\end{theorem}


\begin{proof}[Proof of \refT{T1}]
Fix $r\ge0$. We let $c_1, c_2,c_3$ denote 
unimportant positive constants that may
depend on $r$, but not on $n$.

We use \refT{T58}, with 
$Y_{n,C}:=X_C$ for
$C\in\cA_n:=\cC(n,r)$, so $\SW_n=X_{n,r}$.
We use \refL{LgG} and take the dependency graph $\gG_n=\gG(n,r)$.
Then
\begin{align}\label{swN}
  N_n=|\cC(n,r)|=\binom{n}{r}2^{n-r} \le n^r 2^n,
\end{align}
and, by \eqref{thvar} (where $i=0$ yields the leading term),
\begin{align}
  \gss_n:=\Var(X_{n,r})\ge \cc n^{2r} 2^n.
\end{align}
Furthermore, we may take 
\begin{align}
  A_n:=1.
\end{align}
Moreover, from the definition of the graph $\gG(n,r)$ follows that it is
regular, with degree
\begin{align}\label{swM}
  M_n \le 2^r\binom{n}{r}\le\cc n^r.
\end{align}
We choose $m=3$ in \refT{T58}. (This is the standard case, which corresponds
to the method of moment without extra arguments, see the proof in
\cite{SJ58}.)
Then \eqref{swN}--\eqref{swM} yield
\begin{align}
(N_n/M_n)^{1/m} M_n A_n/\gs_n 
=N_n^{1/3}M_n^{2/3}A_n/\gs_n
\le \cc\frac{ n^r 2^{n/3}}{n^r 2^{n/2}}
\to0
\end{align}
as \ntoo, so \eqref{t58a} holds. Consequently, \refT{T58} applies and shows
\refT{T1}.
\end{proof}


\section{Conclusion }\label{S:Conc}

In the first part, we demonstrated  the power of computers to {\it automatically} derive complicated explicit expressions, all polynomials in $n$ and $2^n$, for low moments of the
random variable {\it number of fixed-size implicants} of a Boolean function of $n$ variable, or equivalently, the number of fixed-dimensional subcubes of
a random subset of the $n$-dimensional unit cube. These implied that the scaled first few moments for low-dimensional subcubes
tend, as $n$ goes to infinity, to those of the Normal Distribution, suggesting, but by no means proving, that for each fixed dimension, this random variable 
is {\it asymptotically normal}.

In the second part, we demonstrated the power of {\it human ingenuity}, and
the traditional lore of probability and statistics, by presenting a simple proof
of asymptotic normality, first proved in \cite{UK}.


\def\nobibitem#1\par{}



\end{document}

